
 Trouble with PB4. With the upper 1/2 band off, I could not keep the TWTs on (this is just our diode 
problem) until I removed about 6dB of gain. However, we cool the beam thinner with only the lower 
1/2 band on than with only the upper 1/2 band or both 1/2 bands on. Further, with either one of the 
upper 1/2 band low-level amps off, the beam gets thinner. 

 We are set up to run with D:PV4AU2 off. I have also modified the screen image for P34 to warn 
Operators not to lower the PB4 attenuator below 14.0 dB. I will fill out a work order to investigate the 
problems with PB4. 

 After work, beam with was narrower longitudinally coming out of the Debuncher, as evidenced by 
having to lower ARF1.

○ Took extra time during the day shift to investigate our Debuncher Momentum cooling issues

 Phil replaced the power supply
 Can only keep the ion pump on if the control cable from the CIA crate is disconnected.
 Controls experts will need to take down the CIA crate to fix this.

○ D:IP202 is off.

○

 Started tripping.
 Checked out DRF1-6, PA current was very low (less than 5mA instead of the normal 10mA). This could 

have been the main cause for the trips. To give it a little rest, the output was turned down to 827kV 
(from 850kV). If it remains stable for a day or two, we can turn it back up. The sum is now 5.12MV.

DRF1-6

 Coincidentlally happened with stacktail TWT #14 was turned on.  That was not the cause
 Pbars were lost when D:ELAM was turned back on.  There is a large tune excursion when D:ELAM is 

turned back on (field free region is not actually field free).   This can be avoided b y knobbing up the 
output.

○ After owl shift glitch, lost 4ma

○

 <stacking rate> =  25.1ma/hr
 <production> =  19.5 pbar/Mp
 <beam on target> = 7.9 Tp

The Stacking Numbers

Stacking

○ Unstacked 500mA in 60 transfers over 20 sets with an average efficiency of 95.7%.

Transfers

Further work on Debuncher momentum cooling○

○ Stacktail Phasing with 30e10 - looking for opportunistic non-stacking time
Stacktail tank moving - parasitic○

Jim Morgan would like to change beamline C204 limits based on calculations using the model.  We will be 
doing one plane of one beamline at a time.

○

Studies

 Pbars stacked: 506.58 E10
 Time stacking: 23.79 Hr
 Average stacking rate: 21.29 E10/Hr

○ Stacking

 Number of pulses while in stacking mode: 36209
 Number of pulses with beam: 33917
 Fraction of up pulses was: 93.67%

○ Uptime

 Corrected time stacking: 22.28 Hr

○ The uptime's effect on the stacking numbers

The Numbers
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 Corrected time stacking: 22.28 Hr
 Possible average stacking rate: 22.73 E10/Hr
 Could have stacked: 540.81 E10/Hr

○ Recycler Transfers
Pbars sent to the Recycler: 490.73 E10

Number of transfers : 60

Number of transfer sets: 20

Average Number of transfer per set: 3.00

Time taken to shoot including reverse proton tuneup: 00.21 Hr

Transfer efficiency: 95.57%

 Average POT : 7.98 E12
 Average production: 18.73 pbars/E6 protons

○ Other Info

○

Elog Notes 

The Plots
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Tuesday Dayshift
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